PEER REVIEW HISTORY

BMJ Paediatrics Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

ARTICLE DETAILS

TITLE (PROVISIONAL)	Designing and evaluating a novel mobile application for Helping Babies Breathe skills retention in Uganda – comparative study protocol
AUTHORS	Merali, Hasan; Chan, Natalie; Mistry, Niraj; Kealey, Ryan; Campbell, Douglas; Morris, Shaun; Data, Santorino

No Reviews for this paper